

**Author's response to reviews**

**Title:** Prospective study of avian influenza transmission to humans in Egypt

**Authors:**

Ghazi Kayali (ghazi.kayali@stjude.org)
Richard J Webby (Richard.webby@stjude.org)
Xiaoping Xiong (Xiaoping.xong@stjude.org)
Lobna S Sherif (lobnasherif2009@hotmail.com)
Esmat A El-Ghafar (esmat_nrc@yahoo.com)
Mohamed A Ali (mohamedahmedali2004@yahoo.com)

**Version:** 3 **Date:** 9 November 2010

**Author's response to reviews:**

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is "First name Initial Last name" (e.g. Joe F Bloggs).

Changes done.

Authors' contributions: Please indicate that all authors read and approved the final manuscript.

Changes done.

Affiliations: Please remember to include the country for all authors' affiliations.

Changes done.